RESEARCH ARTICLE

BMC Medicine



Fotagliptin monotherapy with alogliptin as an active comparator in patients with uncontrolled type 2 diabetes mellitus: a randomized, multicenter, double-blind, placebo-controlled, phase 3 trial



Mingtong Xu¹⁺, Kan Sun¹⁺, Wenjie Xu²⁺, Chuan Wang¹, Dewen Yan³, Shu Li⁴, Li Cong⁵, Yinzhen Pi⁶, Weihong Song⁷, Qingyuan Sun⁸, Rijun Xiao⁹, Weixia Peng¹⁰, Jianping Wang¹¹, Hui Peng¹², Yawei Zhang¹³, Peng Duan¹⁴, Meiying Zhang¹⁵, Jianying Liu¹⁶, Qingmei Huang¹⁷, Xuefeng Li¹⁸, Yan Bao¹⁹, Tianshu Zeng²⁰, Kun Wang²¹, Li Qin²², Chaoming Wu²³, Chunying Deng²⁴, Chenghu Huang²⁵, Shuang Yan²⁶, Wei Zhang²⁷, Meizi Li²⁸, Li Sun²⁹, Yanjun Wang³⁰, HongMei Li³¹, Guang Wang³², Shuguang Pang³³, Xianling Zheng³⁴, Haifang Wang³⁵, Fujun Wang³⁶, Xiuhai Su³⁷, Yujin Ma³⁸, Wei Zhang³⁹, Ziling Li⁴⁰, Zuoling Xie⁴¹, Ning Xu⁴², Lin Ni⁴³, Li Zhang⁴⁴, Xiangqun Deng⁴⁵, Tianrong Pan⁴⁶, Qijuan Dong⁴⁷, Xiaohong Wu⁴⁸, Xingping Shen⁴⁹, Xin Zhang⁵⁰, Qijing Zou⁵¹, Chengxia Jiang⁵², Jue Xi⁵³, Jianhua Ma⁵⁴, Jingchao Sun² and Li Yan^{1*}

Abstract

Background Dipeptidyl peptidase-4 inhibitors (DPP-4i) have become firmly established in treatment algorithms and national guidelines for improving glycemic control in type 2 diabetes mellitus (T2DM). To report the findings from a multicenter, randomized, double-blind, placebo-controlled phase 3 clinical trial, which was designed to assess the efficacy and safety of a novel DPP-4 inhibitor fotagliptin in treatment-naive patients with T2DM.

Methods Patients with T2DM were randomized to receive fotagliptin (n = 230), alogliptin (n = 113) or placebo (n = 115) at a 2:1:1 ratio for 24 weeks of double-blind treatment period, followed by an open-label treatment period, making up a total of 52 weeks. The primary efficacy endpoint was to determine the superiority of fotagliptin over placebo in the change of HbA1c from baseline to Week 24. All serious or significant adverse events were recorded.

Results After 24 weeks, mean decreases in HbA1c from baseline were -0.70% for fotagliptin, -0.72% for alogliptin and -0.26% for placebo. Estimated mean treatment differences in HbA1c were -0.44% (95% confidence interval [CI]: -0.62% to -0.27%) for fotagliptin versus placebo, and -0.46% (95% CI: -0.67% to -0.26%) for alogliptin versus placebo, and 0.02% (95%CI: -0.16% to 0.19%; upper limit of 95%CI < margin of 0.4%) for fotagliptin versus alogliptin. So fotagliptin was non-inferior to alogliptin. Compared with subjects with placebo (15.5%), significantly more patients

[†]Mingtong Xu, Kan Sun and Wenjie Xu contributed equally to this work.

*Correspondence: Li Yan hfxyl@163.net Full list of author information is available at the end of the article



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

with fotagliptin (37.0%) and alogliptin (35.5%) achieved HbA1c < 7.0% after 24 weeks of treatment. During the whole 52 weeks of treatment, the overall incidence of hypoglycemia was low for both of the fotagliptin and alogliptin groups (1.0% each). No drug-related serious adverse events were observed in any treatment group.

Conclusions In summary, the study demonstrated improvement in glycemic control and a favorable safety profile for fotagliptin in treatment-naive patients with T2DM.

Trial registration ClinicalTrail.gov NCT05782192.

Keywords Dipeptidyl peptidase-4 inhibitors, Fotagliptin, Type 2 diabetes mellitus, HbA1c

Background

The prevalence of type 2 diabetes mellitus (T2DM) has increased markedly with an estimated number of 347 million individuals worldwide in 2008, and forecasted to increase to 7079 individuals per 100,000 by 2030 [1]. For T2DM management, a lot of guidelines have recommended metformin as a first-line therapy in combination with healthy diet and exercise, and provided recommendations on second-line therapies when metformin is unable to achieve or maintain long-term glycemic control [2-4]. However, the selection of second-line therapies for T2DM is challenging. Although algorithms provide evidence-based principles and guidelines, several factors should be taken into consideration to determine the optimum approach, such as patient age, individual compliance, financial conditions and diabetes complications [5, 6].

Dipeptidyl peptidase-4 inhibitors (DPP-4i) could improve glycemic control by preventing the rapid degradation of incretin hormones and inhibiting glucagon secretion [7]. With a low risk of hypoglycemia and no weight gain, nearly 0.7% reduction in HbA1c was reported when other DPP-4 inhibitors were given either alone or in combination with metformin for the treatment of T2DM patients [8-10]. Fotagliptin (Salubris Pharmaceuticals, Shenzhen, China) is a selective, oncedaily, novel DPP-4 inhibitor approved for glycemic management of T2DM. Preclinical pharmacological studies showed that fotagliptin could inhibit DPP-4 with IC₅₀of 2.27 nM [11]. Fotagliptin was not primary metabolized by cytochrome P450 enzymes. There were 2 major metabolites, M1 had no inhibitory effect on DPP-4, M2-1 had slight inhibition [12]. The efficacy and safety profiles of fotagliptin have been characterized in previous studies. Fotagliptin exhibited favorable pharmacokinetic results as it can achieve high and steady DPP-4 inhibition. Fotagliptin was rapidly absorbed, which t_{max} was obtained at 1-2 h in T2DM patients, also enabling a maximum inhibition of DPP-4 within 1-2 h post administration [12, 13]. Recently, a phase 1 clinical trial has been conducted in fourteen eligible Chinese patients with T2DM, which showed that fotagliptin was safe and well tolerated [12]. In a phase 2 clinical study, patients who were treated with fotagliptin monotherapy of 6 mg, 12 mg and 24 mg once a day for 12 weeks showed significant improvement in the HbA1c control as compared with placebo. The hypoglycemic effect increased with the treatment duration, and the biggest decrease in HbA1c was observed in the 12 mg fotagliptin group.

This is the first phase 3 randomized, double-blind, placebo-controlled clinical study to evaluate the efficacy and long-term safety of fotagliptin in treatment-naive T2DM patients uncontrolled with diet and exercise intervention, comprising 24 weeks of double-blind treatment period followed by an open-label treatment period, making up a total of 52 weeks.

Methods

Study design and randomization

Patients with type 2 diabetes mellitus (T2DM) were randomized to receive fotagliptin, alogliptin or placebo. Subjects who met the inclusion criteria would enter the 4 weeks of placebo run-in period. At the end of the run-in period, a baseline enrollment evaluation was performed. Upon evaluation, all eligible subjects would enter the 24 weeks of double-blind treatment period and were randomized into the fotagliptin group (12 mg once daily) or alogliptin group (25 mg once daily) or placebo group at a 2:1:1 ratio. Randomization and drug dispensation were performed with an interactive web response system (IWRS;eBalance version 5.3.7). A stratified randomization method with the permuted block randomization algorithm was used. The blocks were dynamically allocated to each site and stratum from the randomization list. A unique ID number was provided by the vendor and marked on the medication box. Using central randomization, randomization codes were assigned to eligible participants by the IWRS system based on stratification factors (baseline HbA1c level < 8.5% or > 8.5%) and the block size. After 24 weeks of double-blind treatment, subjects would enter the extended open-label treatment period. Subjects in the placebo group were to be switched to fotagliptin (12 mg once daily) treatment, while patients

in the fotagliptin and alogliptin groups continued the same treatment until the end of the whole 52 weeks.

The trial was conducted in accordance with the *Declaration of Helsinki* and *Good Clinical Practice* principles. The protocol was approved by the independent institutional review boards or ethics committees of each site that participated in the study. All investigational medicinal products (IMP) and matching placebo were provided by Shenzhen Salubris Pharmaceuticals Co., Ltd. The written informed consent was obtained from all the patients before the implementation of study procedures. The study was registered on ClinicalTrails. gov, number NCT05782192.

Population

Subjects who were 18–75 years old and with previously untreated T2DM (no oral or injected anti-diabetes treatment before 8 weeks of randomisation) were eligible for screening. After a 4-week diet and exercise run-in period, eligible study participants with poor glycemic control (hemoglobin A1c [HbA1c] values of 7.5% to 10.5% inclusive and fasting blood glucose [FBG] \leq 13.9 mmol/L) were included in the trail. Main exclusion criteria were: treatment with any antihyperglycaemic medication within the run-in period of the trial; hypoglycaemic unawareness or recurrent severe hypoglycaemia; anaphylactic reaction or contraindication to any IMP or placebo; impaired renal or hepatic function; acute or severe chronic complications of diabetes. The full inclusion and exclusion criteria are avaliable in Supplementary table 1.

Outcomes and assessments

The primary efficacy endpoint was to evaluate the HbA1c from baseline to Week 24 in T2DM patients treated with fotagliptin 12 mg/day, in parallel control with alogliptin and placebo. The secondary endpoints were change in HbA1c from baseline to Week 52; change in FBG from baseline to Week 52; occurrence of hypoglycemia from baseline to Week 52; change in bodyweight from baseline to Week 52. All adverse events were followed up by investigators and have been sufficiently characterised. The following adverse events were to be reported: serious adverse events occured in more than 2% of the patients in either treatment group; any adverse events group; any other safety events of special interest.

Statistical analysis

The primary objective was to determine the superiority of fotagliptin over placebo in the change of HbA1c from baseline to Week 24, and the two-sided test of superiority/non-inferiority design was adopted with α =0.05 and β =0.20. According to the phase 3 clinical trial results

of other drugs in the same class, it is estimated that the primary efficacy endpoint of HbA1c difference between fotagliptin and placebo is 0.45% [14], assuming a standard deviation of 1.0%. The non-inferiority limit between fotagliptin and alogliptin is 0.4. With the 2:1:1 ratio of fotagliptin group, alogliptin group and placebo group, the sample size of the three groups was estimated to be 160, 80 and 80 patients, respectively. Considering the rate of loss to follow-up and to obtain safety data through Week 52, the sample size was estimated to be 224, 112 and 112 patients in the fotagliptin group, alogliptin group, alogliptin group and placebo group, respectively.

The primary analysis was performed in the full analysis set (FAS), which received at least one dose of the study drug and had at least one posttreatment measurement of the endpoint during the double-blind treatment period. Missing data were imputed using the last observation carried forward (LOCF). HbA1c values at follow-up visits of subjects received rescue were treated as missing values. As a dependent variable, the change of HbA1c from baseline to Week 24 was analyzed using the analysis of covariance (ANCOVA) model. With the treatment group as the fixed effect and the baseline HbA1c as a linear covariate, we assessed the least squares mean and standard error of the primary efficacy endpoint of each treatment group, as well as the differences of the primary efficacy endpoint and its standard error or 95% confidence interval between groups.

Descriptive statistics (frequency and percentage) were used to summarize the patient demographics, incidence of adverse events and hypoglycemic episodes by treatment group. Continuous variables, such as physical examination and clinical laboratory evaluations, were summarized by means \pm standard deviation (SD) for normally distributed data or medians (interquartile ranges) for non-normally distributed data. All statistical tests were two-sided, with *P* value < 0.05 considered as statistically significant. Statistical analysis was performed using SAS version 9.4 (SAS Institute Inc, Cary, NC, USA).

Results

In all, a total of 836 patients with T2DM from 56 sites were screened for eligibility. After the run-in period, 458 patients were randomized in the double-blind period as follows: 230 patients in the fotagliptin group, 113 patients in the alogliptin group, and 115 patients in the placebo group. In the fotagliptin group, alogliptin group and placebo group, 19 (8.2%), 14 (12.3%), and 9 (7.8%) patients prematurely discontinued the study and the most common reason for study discontinuation was voluntary withdrawal (9, 10, and 9 patients, respectively). There were 91% (416/458) patients compliant with the study drug dosing throughout the double-blind period while 94% (389/416) of these patients completed the whole 52-week treatment period. Details of the patients' disposition are showed in Fig. 1. The FAS included 450 patients. At baseline, demographic and clinical characteristics were well balanced across treatment groups. There were no differences across the fotagliptin, alogliptin and placebo groups in mean HbA1c and other glycemic indicators (Table 1).

The primary efficacy endpoint was change from baseline in HbA1c at Week 24. As shown in Fig. 2A, HbA1c reductions were superior with fotagliptin and alogliptin versus placebo at Week 24 (All *P* values < 0.0001). Mean decreases in HbA1c from baseline to Week 24 were -0.70% for fotagliptin, -0.72% for alogliptin and -0.26% for placebo (Fig. 2B). Estimated mean treatment differences were -0.44% (95% CI: -0.62% to -0.27%) for fotagliptin versus placebo, and -0.46% (95% CI: -0.67% to -0.26%) for alogliptin versus placebo at Week 24. Fotagliptin was also non-inferior to alogliptin, as estimated mean treatment difference of fotagliptin vs alogliptin was 0.02% (95%CI: -0.16% to 0.19%; upper limit of 95%CI < margin of 0.4%). Results from sensitivity analysis supported the results of confirmatory analysis (Supplementary Figs. 2, 3, 4). A total of 20 (4.4%) subjects used rescue therapy with biguanides in the double-blind period, including 5 patients (2.2%) in the fotagliptin group, 6 patients (5.5%) in the alogliptin group, and 9 patients (8.0%) in the placebo group. In the extended treatment period, 92% (97/106) subjects in the placebo group used fotagliptin at a dose of 12 mg once daily and there were no differences in HbA1c from baseline to Week 52 across treatment groups (Fig. 2A).

Compared with subjects with placebo, significantly more patients with fotagliptin and alogliptin achieved the HbA1c targets (<7.0% and \leq 6.5%) after 24 weeks of treatment: 20.7%, 20.0% and 4.4% patients achieved HbA1c \leq 6.5% in the fotagliptin group, alogliptin group and placebo group, respectively; 37.0%, 35.5% and 15.5% patients achieved HbA1c <7.0% in the fotagliptin group, alogliptin group and placebo group, respectively (Fig. 3A). The proportion of patients achieved the HbA1c targets was similar between fotagliptin and alogliptin treatment groups at Week 24.

In contrast to placebo, fotagliptin and alogliptin resulted in significantly greater decrease in mean FBG after 24 weeks of treatment (Fig. 3B). Mean weight loss



Fig. 1 Study patient disposition

	Fotagliptin group (n=227)	Alogliptin group (n = 110)	Placebo group (n = 113)	Р
Age (years)	53.0 (9.03)	53.7 (10.37)	54.2 (9.75)	0.4508
Male [n (%)]	140 (61.7)	70 (63.6)	72 (63.7)	0.9080
BMI (kg/m2)	25.67 (3.12)	25.69 (3.07)	25.65 (2.91)	0.9911
Weight (kg)	70.77 (11.18)	69.76 (11.69)	69.20 (11.45)	0.2494
SBP (mmHg)	124.9 (11.6)	126.4 (12.5)	125.3 (12.0)	0.5502
DBP (mmHg)	81.1 (7.9)	81.9 (8.0)	81.0 (7.4)	0.7377
eGFR (ml/min per 1.73 m2)	104.92 (11.45)	101.84 (14.19)	102.18 (13.98)	0.0276
ALT (U/L)	23.0 (17.0, 33.0)	21.9 (15.0, 33.0)	24.0 (16.0, 35.3)	0.3979
AST (U/L)	20.0 (16.3, 25.7)	21.0 (16.0, 27.0)	21.0 (17.0, 27.0)	0.5847
Duration of diabetes (months)	11.00 (1.80,36.50)	13.20 (2.70,38.60)	19.25 (2.35,34.85)	0.6666
FPG (mmol/L)	9.34 (1.79)	9.40 (2.06)	9.47 (1.79)	0.8566
HbA1c (%)	8.08 (0.73)	8.10 (0.74)	8.12 (0.70)	0.8495
Fasting insulin (µIU/ml)	10.67 (7.55,15.80)	10.36 (6.61,16.32)	10.58 (7.19,15.04)	0.7627
Fasting C-peptide (ng/ml)	2.49 (2.01,3.14)	2.37 (1.96,3.07)	2.51 (1.93,3.27)	0.5884
ΗΟΜΑ-β	35.62 (23.40,53.77)	33.14 (20.46,57.54)	32.91 (22.89,48.21)	0.4523
HOMA-IR	4.45 (3.10,6.93)	4.61 (2.66,6.41)	4.47 (2.94,7.03)	0.8199

 Table 1
 Demographic and clinical characteristics of the patients at baseline

1 Data were means ± SD or medians (interquartile ranges) for skewed variables or numbers (proportions) for categorical variables

2 P for trend was calculated for the linear regression analysis tests across the groups. P values were for the ANOVA or $\chi 2$ analyses across the groups

3 BMI body mass index, SBP systolic blood pressure, DBP diastolic blood pressure, eGFR estimated glomerular filtration rate, FPG fasting plasma glucose, y-GGT y-glutamyltransferase



Fig. 2 Differences in primary clinical end points (Full-analysis-set). A HbA1c values. B HbA1c values change in double-blind stage. Data are mean and error bars are SEs. The ETD and corresponding 95% CI were estimated using an ANCOVA in the FAS. Last observation carried forward imputation was used for missing values

did not change substantially during the whole 52 weeks of treatment and no more than 1 kg of weight change was found in any group (Fig. 3C). During the whole 52 weeks of treatment, the overall incidence of hypoglycemia was relatively low (1.0% [2/211] for fotagliptin group, 1.0% [1/99] for alogliptin group and 3.8% [4/106] for placebo group).

Changes in additional efficacy measures from baseline to Week 24 and Week 52 are provided in Table 2. The assessment of β -cell function from baseline to Week 24



Fig. 3 Differences in secondary endpoints. (Full-analysis-set). A Proportion of participants achieving HbA1c target ≤ 6.5% in double-blind stage. and proportion of participants achieving HbA1c target < 7.0% in double-blind stage. B FBG change in double-blind stage. Data are mean and error bars are SEs. C FBG change in double-blind stage. Data are mean and error bars are SEs.

showed that both fotagliptin and alogliptin were associated with significant improvements in homoeostasis model assessment of β -cell (HOMA- β) compared with baseline, but no treatment related differences were recorded for HOMA index of insulin resistance (HOMA-IR), or fasting insulin concentration. Meanwhile, fotagliptin was associated with significant improvements in fasting C-peptide concentration compared with baseline. No significant changes in the aminotransferase and lipid profiles were observed in the three groups. DPP-4i may have a small effect on exocrine gland of pancreas as the serum amylase and lipase levels showed a minor increase across the fotagliptin, alogliptin and placebo groups from baseline to Week 52. Moreover, serum creatinine increased slightly in fotagliptin group.

Table 3 summarized the clinical adverse events, which showed that fotagliptin and alogliptin were well tolerated in all treatment groups. No deaths were reported during the study in all treatment groups. No serious adverse events occurred in more than 2% of the patients in either treatment group. All adverse events occurred in 4% of participants or fewer, and adverse events occurred infrequently in all groups in double-blind period (3.1% for fotagliptin group, 3.6% for alogliptin group and 6.2% for placebo group). Adverse events that were considered to be treatment-emergent (TEAEs) and occurred in more than 5% of any treatment group are shown in Table 3. The most commonly reported TEAE was lipase increased, reported by 7 (3.1%), 10 (9.1%), 2 (1.8%) of patients in the fotagliptin, alogliptin and placebo groups, respectively. Statistically significant difference in the occurrence of lipase increased was observed between the alogliptin and placebo groups. Minor hypoglycemia was reported in participants treated with fotagliptin (2 [0.9%]) and alogliptin (0 [0.0%]) in double-blind period. The occurrence of hypoglycemia was similar in all treatment groups in the extended period. Unexpectedly, seven patients in the placebo group experienced hypoglycemia in doubleblind period. All hypoglycemia events were considered mild to moderate in intensity with no need for assistance from others.

Discussion

The multicenter clinical study showed that in T2DM patients inadequately controlled with diet and exercise intervention, fotagliptin 12 mg once daily for 24 weeks provided superior glycemic control compared with placebo, as assessed by reductions in HbA1c and FBG. No clinically significant difference in the improvements in clinical response of glycemic control was observed between the fotagliptin group and the alogliptin group.

HbA1c represents the most powerful predictor of diabetes related outcomes and mortality [15, 16]. The phase 3 randomized, double-blind, placebo-controlled clinical study of fotagliptin achieved its primary efficacy endpoint. In treatment-naive T2DM patients uncontrolled with diet and exercise intervention, fotagliptin monotherapy achieved clinically and significant meaningful amelioration in the percentage of patients reached HbA1c targets (<7.0% and \leq 6.5%). Current guidelines suggest that should

Table 2 Change in additional secondary measures of efficacy from baseline to week 24 and week 52

	Mean change from baseli	ne to week 24			
	Fotagliptin group	Alogliptin group	Placebo group	Р	
Fasting insulin (µIU/ml)	0.73(-0.13,1.59)	0.44(-0.51,1.38)	-0.11(-1.18,0.96)	0.671	
Fasting C-peptide (pmol/L)	0.13(0.03,0.23)	0.08(-0.03,0.19)	-0.05(-0.16,0.06)	0.146	
ΗΟΜΑ-β	11.00(7.46,14.55)	7.21(3.39,11.02)	4.61(0.66,8.56)	0.066	
HOMA-IR	-0.12(-0.58,0.33)	-0.16(-0.63,0.30)	-0.46(-0.99,0.06)	0.980	
ALT (U/L)	-2.26(-3.64,-0.89)	-2.31(-4.17,-0.44)	4.82(-5.99,15.63)	0.820	
AST (U/L)	-1.01(-1.83,-0.19)	-0.81(-2.10,0.48)	1.94(-3.79,7.68)	0.836	
Serum creatinine (umol/L)	2.28(1.19,3.37)	-0.50(-1.96,0.96)	0.76(-0.97,2.49)	0.018	
eGFR (ml/min per 1.73 m2)	-1.83(-2.85,-0.80)	0.90(-0.39,2.19)	-0.96(-2.64,0.73)	0.018	
LDL-C (mmol/L)	0.02(-0.07,0.11)	-0.04(-0.16,0.09)	0.06(-0.07,0.19)	0.225	
TG (mmol/L)	0.15(-0.08,0.37)	0.23(0.01,0.44)	-0.12(-0.28,0.04)	0.261	
Serum amylase (U/L)	4.10(1.89,6.30)	1.92(-1.36,5.20)	0.52(-2.27,3.32)	0.107	
Serum lipase (U/L)	6.35(2.02,10.67)	6.14(3.03,9.25)	0.67(-1.29,2.63)	0.003	
	Mean change from baseline to week 52				
	Fotagliptin group	Alogliptin group	Control group	Р	
Fasting insulin (µIU/mI)	-0.52(-1.29,0.26)	0.24(-0.95,1.43)	0.16(-1.12,1.44)	0.661	
Fasting C-peptide (pmol/L)	0.07(-0.01,0.15)	0.14(-0.00,0.28)	0.14(-0.02,0.29)	0.640	
ΗΟΜΑ-β	3.76(0.56,6.96)	4.42(-0.08,8.93)	6.94(1.96,11.92)	0.648	
HOMA-IR	-0.44(-0.91,0.04)	-0.20(-0.79,0.40)	-0.36(-1.03,0.31)	0.630	
ALT (U/L)	-0.37(-2.15,1.41)	-1.25(-3.35,0.86)	2.28(-1.59,6.14)	0.697	
AST (U/L)	0.22(-0.75,1.19)	-0.53(-1.77,0.71)	1.77(-0.22,3.76)	0.443	
Serum creatinine (umol/L)	3.15(1.60,4.71)	0.67(-1.17,2.51)	2.66(0.84,4.48)	0.144	
eGFR (ml/min per 1.73 m ²)	-2.34(-3.54,-1.14)	-0.62(-2.40,1.15)	-2.24(-3.88,-0.59)	0.184	
LDL-C (mmol/L)	0.03(-0.07,0.13)	-0.04(-0.18,0.11)	0.04(-0.11,0.20)	0.676	
TG (mmol/L)	0.27(-0.05,0.58)	0.22(0.02,0.42)	-0.06(-0.32,0.21)	0.111	
Serum amylase (U/L)	4.59(3.07,6.10)	3.16(0.79,5.54)	5.62(2.51,8.72)	0.744	
Serum lipase (U/L)	6.83(4.30,9.35)	6.37(3.56,9.18)	6.65(3.90,9.41)	0.113	

be set for most adults with diabetes, as concerns remain that aggressive hypoglycemic treatment may increase the risk of diabetic complications [17, 18]. Fotagliptin demonstrated a favorable safety profile during the 52 weeks treatment, and there were no serious hypoglycemia or severe adverse events that required medical assistance. Clinically significant hypoglycemia occurred only in two patients (0.9%) treated with fotagliptin over 24 weeks, and the events were mild in nature. The incidence of hypoglycemia reported in this study is consistent with other studies to investigate the efficacy and safety of DPP-4 inhibitors therapy [19, 20]. Mild but not clinically significant weight change was observed in all treatment groups, including placebo. These may be attributed to the pharmacokinetic and pharmacodynamic results of both fotagliptin and alogliptin. One recent study in Chinese patients with T2DM indicated that fotagliptin could increase plasma GLP-1 concentration while DPP-4 inhibition was continuously maintained at a steady state [12]. Still, relevant results were based on the design of the study and patients were closely monitored to ensure that they followed the diet and exercise guidelines of the trail. Given an increasing use of DPP-4 inhibitors to treat patients with T2DM in the real-world setting, potential pancreatitis risk has been of concern because of its consequent pharmacological mechanism of the pancreas [7, 21]. Overall, no pancreatitis occurred during the whole treatment period. As expected, slight increases of amylase and lipase levels were observed with fotagliptin and alogliptin from Week 4 to Week 24, and this trend gradually decreased in the later period of the trail. No significant change from baseline in pancreatic enzymes was observed in placebo group.

Fotagliptin is a novel highly selective DPP-4 inhibitor under clinical development for the increased levels of intact bioactive GIP and GLP-1. Based on previous studies, fotagliptin could increase active GLP-1 concentrations and have no obvious influence on DPP-8 and DPP-9, thus making it safer to treat T2DM [13]. To our knowledge, this was the first adequately powered trial to investigate the efficacy and safety profiles of fotagliptin in patients with uncontrolled T2DM. The reductions in HbA1c with fotagliptin were much the same as

Page 8 of	10

		Fotagliptin group	Alogliptin group	Placebo group	Р
Serious adverse events th	at were present in more than 2% of the	patients in either trial grou	up, n (%)		
Double-blind stage	-	-	-	-	-
Extended stage	-	-	-	-	-
Adverse events that were	considered to be related to therapeutic	drug or placebo in more	than 5% of the patients in	either trial group, n (%)	
Double-blind stage	Lipase increased	7 (3.1)	10 (9.1)#	2 (1.8)	0.0217
Extended stage	-	-	-	-	-
Any other safety events o	f special interest, n (%)				
Double-blind stage	Hypoglycemia	2 (0.9)*	0 (0.0)#	7 (6.2)	0.0038
	Myocardial ischaemia	2 (0.9)	2 (1.8)	2 (1.8)	0.6441
	Arteriosclerosis coronary artery	2 (0.9)	0 (0.0)	1 (0.9)	1.0000
	Acute coronary syndrome	1 (0.4)	0 (0.0)	0 (0.0)	1.0000
Extended stage	Hypoglycemia	2 (1.0)	1 (1.0)	4 (3.8)	0.1604
	Myocardial ischaemia	2 (1.0)	0 (0.0)	2 (1.9)	0.5587
	Angina pectoris	1 (0.5)	0 (0.0)	0 (0.0)	1.0000
	Arteriosclerosis coronary artery	1 (0.5)	0 (0.0)	1 (1.0)	1.000
Any adverse events more	than 5% in either trial group				
Double-blind stage	Dyslipidemia	28 (12.3)	17 (15.5)	15 (13.3)	0.7600
	Hyperuricemia	18 (7.9)	7 (6.4)	3 (2.7)	0.1622
	Urinary tract infection	13 (5.7)	9 (8.2)	7 (6.2)	0.7051
	Upper respiratory tract infection	14 (6.1)	8 (7.3)	6 (5.3)	0.8416
	Lipase increased	8 (3.5)	11 (10.0)##	2 (1.8)	0.0081
	Hypoglycemia	2 (0.9)*	0 (0.0)#	7 (6.2)	0.0038
Extended stage	Dyslipidemia	17 (8.2)	10 (10.2)	5 (4.8)	0.3880
	Hyperuricemia	4 (1.9)	7 (7.1)	7 (6.7)	0.0347
	Abnormal hepatic function	6 (2.9)	4 (4.1)	6 (5.8)	0.4256

Table 3 Summary of clinical adverse events in the treated set of safety population

1 The safety population was defined as patients who took at least one dose of therapeutic drug or placebo

2 Data are number (% of total participants in treatment group)

3 *P < 0.05 compared with control group; ${}^{\#}P$ < 0.05; ${}^{\#\#}P$ < 0.01 compared with control group

4 This table lists serious adverse events that were present in more than 2% of the patients in either trial group, adverse events that were considered to be related to therapeutic drug or placebo in more than 5% of the patients in either trial group, and any other safety events of special interest

those with alogliptin in the double-blind period. In addition, the study results did not demonstrate any benefit of fotagliptin for most additional secondary efficacy measures except for fasting C-peptide, which we suspected that may be related to the relieved glycotoxicity in some patients. Serum creatinine increased slightly with fotagliptin and differences were small but significant for three groups in the double-blind period. Nevertheless, such difference no longer existed in the extended treatment period. In the double-blind period, the proportion of subjects receiving rescue therapy in fotagliptin group was significantly lower than that in the placebo group, but the proportion increased in the extended period for both the fotagliptin group and alogliptin group. Interestingly, after the conversion to fotagliptin in the extended period, the proportion of subjects receiving rescue therapy in the original placebo group increased, although only 24 weeks of delayed hypoglycemic treatment. Therefore, in order to obtain long-term benefits, it is reasonable for patients with T2DM to start hypoglycemic therapy as soon as possible after the diagnosis of diabetes.

However, limitations existed in the present trial. First of all, there is a potential limitation in interpreting effect of fotagliptin compared with placebo on risk of major cardiovascular (CV) events, as the study duration may be too short to modulate CV related clinical outcomes. Four previous CV outcome trials of DPP-4 inhibitors have demonstrated a non-inferior risk of a composite CV outcome when compared with placebo [22–25]. Secondly, the follow-up duration was relatively short at approximately 52 weeks. No other benefits and risks of long-term treatment with fotagliptin were evaluated. Third, although the comparison between fotagliptin and placebo in previously untreated diabetes patients who are not controlled with diet and exercise provided the exact glucose-lowering of the efficacy and safety of fotagliptin, we should notice that treatment with metformin is still the first-line therapy for patients with T2DM [26].

Conclusion

Overall, the trial demonstrated improvement in glycemic control for fotagliptin monotherapy with 12 mg once daily in previously untreated T2DM patients uncontrolled with lifestyle intervention. Furthermore, fotagliptin treatment was not associated with greater risk of hypoglycemia episodes and weight gain, as compared with placebo and alogliptin. Thus, fotagliptin is a potential new approach for the treatment of T2DM patients who are inadequately controlled with diet and exercise intervention.

Abbreviations

ANCOVA	Analysis of covariance model
CV	Cardiovascular
DPP-4i	Dipeptidyl peptidase-4 inhibitors
etd	Estimated treatment difference
FBG	Fasting blood glucose
HbA1c	Hemoglobin A1C
ΗΟΜΑ-β	Homoeostasis model assessment of $\beta\mbox{-cell}$
HOMA-IR	Homoeostasis index of insulin resistance
IC50	Half of maximal inhibitory concentration
IWRS	Interactive web response system
IMP	Investigational medicinal products
GLP-1	Glucagon-like peptide-1
LOCF	The last observation carried forward
T2DM	Type 2 diabetes mellitus
t _{max}	Maximum serum concentration
TEAE	Treatment-emergent adverse event
SD	Standard deviation
SE	Standard error
95% CI	95% Confidence interval

Supplementary Information

The online version contains supplementary material available at https://doi.org/10.1186/s12916-023-03089-x.

Additional file 1: Supplementary Table 1. Full inclusion and exclusion criteria. Supplementary Figure 1. Clinical study design. Supplementary Figure 2. Differences in primary clinical end points (Full-analysis-set; Center effect) HbA1c values change in double-blind stage. Data are mean and error bars are SEs. The ETD and corresponding 95% CI were estimated using an ANCOVA with center effect in the FAS. Last observation carried forward imputation was used for missing values. Supplementary Figure 3. Differences in primary clinical end points (Full-analysis-set; Observed data) HbA1c values change in double-blind stage. Data are mean and error bars are SEs. The ETD and corresponding 95% CI were estimated using an ANCOVA without missing-value imputation in the FAS, including observation data after rescue. Supplementary Figure 4. Differences in primary clinical end points (Full-analysis-set; MMRM) HbA1c values change in double-blind stage. Data are mean and error bars are SEs. The ETD and corresponding 95% CI were estimated using a mixed model of repeated measure (MMRM) without missing-value imputation in the FAS. Supplementary Figure 5. The multiplicity strategy for primary end point was the fixed-sequence test.

Acknowledgements

We are indebted to the participants in the present study for their outstanding support and to our colleagues for their valuable assistance. Medical writing

and editorial support were provided by Xiangnan Zhou (Clinical Operation Director at Shenzhen Salubris Pharmaceuticals), Chongyang Deng (Medical Manager at Shenzhen Salubris Pharmaceuticals) and Yue Yang (Statistician at Shenzhen Salubris Pharmaceuticals).

Authors' contributions

Protocol design: LY, MX, KS and WX. Project Administration: WX and JS. Methodology and Resources: all authors. Data analysis: KS and LY. Validation: all authors. Manuscript writing: KS. Review and editing: all authors. The authors read and approved the final manuscript.

Funding

This study was funded by Shenzhen Salubris Pharmaceuticals Co.,Ltd. The study was also funded in part, by grants from the Shenzhen Science and Technology Program for Undertake the National Science and Technology Major Project (CJGJZD20210408091600001).

Availability of data and materials

The data that support the findings of this study are available on request from the corresponding authors after the completion of the study. The data are not publicly available due to privacy or ethical restrictions.

Declarations

Ethics approval and consent to participate

The study was approved by the ethics committee at each study site and conducted according to the Declaration of Helsinki, Guidelines for Good Clinical Practice, and local laws and regulations. All patients provided written informed consents.

Consent for publication

All authors believe that the manuscript represents valid work and have reviewed and approved the final version. The work has not been published previously, and is not under consideration for publication elsewhere, in part or in whole.

Competing interests

Wenjie Xu and Jingchao Sun were reported being employed at Shenzhen Salubris Pharmaceuticals. The remaining authors declare no competing interests.

Author details

Sun Yat-Sen Memorial Hospital, Sun Yat-Sen University, Guangzhou, China. ²Shenzhen Salubris Pharmaceuticals Co., Ltd, Shenzhen, China. ³Shenzhen Second People's Hospital, Shenzhen, China.⁴Huizhou Municipal Central Hospital, Huizhou, China. ⁵The Fifth Affiliated Hospital of Sun Yat-Sen University, Zhuhai, China. ⁶The First Hospital of Changsha, Changsha, China. ⁷Chenzhou First People's Hospital, Chenzhou, China.⁸Yueyang Central Hospital, Yueyang, China. ⁹The First Peole's Hospital of Xiangtan City, Xiangtan, China. ¹⁰Yiyang Central Hospital, Yiyang, China. ¹¹The Second Affiliated Hospital of the University of South China, Hengyang, China. ¹²Yichun People's Hospital, The Affiliated Hospital of Yichun University, Yichun, China. ¹³Pingxiang People's Hospital, Pingxiang, China.¹⁴The People's Hospital of Nanchang, The Third Hospital of Nanchang, Nanchang, China. ¹⁵The Second Affiliated Hospital of Nanchang University, Nanchang, China. ¹⁶The First Affiliated Hospital of Nanchang University, Nanchang, China. ¹⁷Xinyu People's Hospital, Xinyu, China. ¹⁸Taihe Hospital, Affilited Hospital of Hubei University of Medicine, Shiyan, China.¹⁹Renmin Hospital of Wuhan University, Wuhan, China.²⁰Union Hospital, Tongji Medical College, Huazhong University of Science and Technology, Wuhan, China.²¹Nanjing Jiangning Hospital, The Affiliated Jiangning Hospital of Nanjing Medical University, Nanjing, China. ²²Chongming Branch, Xinhua Hospital Affiliated to Shanghai Jiaotong University School of Medicine, Shanghai, China. ²³The 2nd School of Medicine, WMU, The 2nd Affiliated Hospital and Yuying Children's Hospital of WMU, Wenzhou, China. ²⁴Zigong Fourth People's Hospital, Zigong, China. ²⁵Bishan Hospital of Chongging, Bishan Hospital of Chongqing Medical University, Chongqing, China.²⁶Fourth affiliated hospital of Harbin Medical University, Harbin, China. ²⁷The First Hospital of Qiqihar, Qiqihar, China.²⁸The Affiliated Hospital of Yanbian University, Yanbian, China. ²⁹Siping Central People's Hospital, Siping, China. ³⁰The Second Norman Bethune Hospital of Jilin University, Jilin, China. ³¹Emergency General

Hospital, Beijing, China. ³²Beijing Chao-Yang Hospital, Capital Medicine University, Beijing, China.³³ Jinan Central Hospital, Central Hospital Affiliated to Shandong First Medical University, Jinan, China. ³⁴Handan Central Hospital, Handan, China. ³⁵Handan First Hospital, Handan, China. ³⁶The Fourth Hospital of He Bei Medical University, Shijiazhuang, China. ³⁷Cangzhou Hospital of Integrated TCM-WM Heibei, Cangzhou, China. ³⁸The First Affiliated Hospital and College of Clinical Medicine, Henan University of Science and Technology, Luoyang, China. ³⁹Puyang Oilfield General Hospital, Puyang, China. ⁴⁰Inner Mongolia Baogang Hospita, Baotou, China. ⁴¹Zhongda Hospital Southeast University, Naniing, China.⁴²The First People's Hospital of Lianyungang, Lianyungang, China. ⁴³The First People's Hospital of Huzhou, The First Affiliated Hospital of Huzhou Teacher College, Huzhou, China.⁴⁴Hainan Third People's Hospital, Sanya, China.⁴⁵Wuhan Third Hospital, Wuhan, China.⁴⁶The Second Hospital of Anhui Medical University, Hefei, China. ⁴⁷People's Hospital of Zhengzhou, People's Hospital of Henan University of Chinese Medicine, Zhengzhou, China. ⁴⁸Zhejiang Provincial People's Hospital, People's Hospital of Hangzhou Medical College, Hangzhou, China. ⁴⁹Zhongshan Hospital Xiamen University, Xiamen, China. ⁵⁰Genertec Liaoyou Gem Flowe Hospital, Panjin, China. ⁵¹The Central Hospital of Yougzhou, Yongzhou, China.⁵²The First People's Hospital of Zunyi, Zunyi, China. ⁵³The Affiliated Hospital of Xuzhou Medical University, Xuzhou, China. ⁵⁴Nanjing First Hospital, Nanjing, China.

Received: 6 April 2023 Accepted: 21 September 2023 Published online: 09 October 2023

References

- Khan MAB, Hashim MJ, King JK, Govender RD, Mustafa H, Al KJ. Epidemiology of type 2 diabetes - global burden of disease and forecasted trends. J Epidemiol Glob Health. 2020;10(1):107–11.
- American DA. Standards of medical care in diabetes-2022 abridged for primary care providers. Clin Diabetes. 2022;40(1):10–38.
- American Diabetes Association Professional Practice C. 9. Pharmacologic approaches to glycemic treatment: standards of medical care in diabetes-2022. Diabetes Care. 2022;45(Suppl 1):S125-S143.
- Davies MJ, D'Alessio DA, Fradkin J, et al. Management of Hyperglycemia in Type 2 Diabetes, 2018. a Consensus Report by the American Diabetes Association (ADA) and the European Association for the Study of Diabetes (EASD). Diabetes Care. 2018;41(12):2669–701.
- Sachinidis A, Nikolic D, Stoian AP, et al. Cardiovascular outcomes trials with incretin-based medications: a critical review of data available on GLP-1 receptor agonists and DPP-4 inhibitors. Metabolism. 2020;111: 154343.
- American Diabetes Association Professional Practice C. 17. Diabetes advocacy: standards of medical care in diabetes-2022. Diabetes Care. 2022;45(Suppl 1):S254-S255.
- Deacon CF. Dipeptidyl peptidase 4 inhibitors in the treatment of type 2 diabetes mellitus. Nat Rev Endocrinol. 2020;16(11):642–53.
- Nagao M, Sasaki J, Sugihara H, et al. Efficacy and safety of sitagliptin treatment in older adults with moderately controlled type 2 diabetes: the STREAM study. Sci Rep. 2023;13(1):134.
- Nomoto H, Takahashi A, Nakamura A, et al. Add-on imeglimin versus metformin dose escalation regarding glycemic control in patients with type 2 diabetes treated with a dipeptidyl peptidase-4 inhibitor plus low-dose metformin: study protocol for a multicenter, prospective, randomized, open-label, parallel-group comparison study (MEGMI study). BMJ Open Diabetes Res Care. 2022;10(6):e002988.
- Chen X, Jiang H, Li H, et al. Saxagliptin combined with additional oral antihyperglycaemic agents in drug-naive diabetic patients with high glycosylated haemoglobin: a 24-week, multicentre, randomized, open-label, active parallel-controlled group clinical trial in China (SUCCESS). Diabetes Obes Metab. 2023;25(1):272–81.
- Wang ZL, Jiang J, Hu P, et al. Development and validation of a UPLC-MS/ MS method for simultaneous determination of fotagliptin and two major metabolites in human plasma and urine. Bioanalysis. 2017;9(4):381–93.
- Wu M, Li QQ, Zhang H, et al. Safety, pharmacokinetics, and pharmacodynamics of a dipeptidyl peptidase-4 inhibitor: a randomized, doubleblinded, placebo-controlled daily administration of fotagliptin benzoate for 14 days for type 2 diabetes mellitus. Clin Pharmacol Drug Dev. 2021;10(6):660–8.

- Ding Y, Zhang H, Li C, et al. Safety and pharmacokinetic interaction between fotagliptin, a dipeptidyl peptidase-4 inhibitor, and metformin in healthy subjects. Expert Opin Drug Metab Toxicol. 2021;17(6):725–31.
- Pan C, Han P, Ji Q, et al. Efficacy and safety of alogliptin in patients with type 2 diabetes mellitus: a multicentre randomized double-blind placebo-controlled phase 3 study in mainland China, Taiwan, and Hong Kong. J Diabetes. 2017;9(4):386–95.
- Wu TE, Su YW, Chen HS. Mean HbA1c and HbA1c variability are associated with differing diabetes-related complications in patients with type 2 diabetes mellitus. Diabetes Res Clin Pract. 2022;192: 110069.
- 16. Ceriello A, Lucisano G, Prattichizzo F, et al. HbA1c variability predicts cardiovascular complications in type 2 diabetes regardless of being at glycemic target. Cardiovasc Diabetol. 2022;21(1):13.
- Buse JB, Wexler DJ, Tsapas A, et al. 2019 Update to: Management of Hyperglycemia in Type 2 Diabetes, 2018. A Consensus Report by the American Diabetes Association (ADA) and the European Association for the Study of Diabetes (EASD). Diabetes Care. 2020;43(2):487–93.
- 18. Tajima A, Tobe K, Eiki JI, et al. Treatment patterns and satisfaction in patients with type 2 diabetes newly initiating oral monotherapy with antidiabetic drugs in Japan: results from the prospective Real-world Observational Study on Patient Outcomes in Diabetes (RESPOND). BMJ Open Diabetes Res Care. 2022;10(6):e003032.
- Chong SC, Sukor N, Robert SA, Ng KF, Kamaruddin NA. Endogenous GLP-1 levels play an important role in determining the efficacy of DPP-IV Inhibitors in both prediabetes and type 2 diabetes. Front Endocrinol (Lausanne). 2022;13:1012412.
- 20. Group GSR, Nathan DM, Lachin JM, et al. Glycemia reduction in type 2 diabetes glycemic outcomes. N Engl J Med. 2022;387(12):1063–74.
- 21. Doni K, Buhn S, Weise A, et al. Safety of dipeptidyl peptidase-4 inhibitors in older adults with type 2 diabetes: a systematic review and meta-analysis of randomized controlled trials. Ther Adv Drug Saf. 2022;13:20420986211072384.
- 22. Scirica BM, Bhatt DL, Braunwald E, et al. Saxagliptin and cardiovascular outcomes in patients with type 2 diabetes mellitus. N Engl J Med. 2013;369(14):1317–26.
- White WB, Cannon CP, Heller SR, et al. Alogliptin after acute coronary syndrome in patients with type 2 diabetes. N Engl J Med. 2013;369(14):1327–35.
- 24. Green JB, Bethel MA, Armstrong PW, et al. Effect of sitagliptin on cardiovascular outcomes in type 2 diabetes. N Engl J Med. 2015;373(3):232–42.
- Rosenstock J, Perkovic V, Johansen OE, et al. Effect of linagliptin vs placebo on major cardiovascular events in adults with type 2 diabetes and high cardiovascular and renal risk: The CARMELINA randomized clinical trial. JAMA. 2019;321(1):69–79.
- Gnesin F, Thuesen ACB, Kahler LKA, Madsbad S, Hemmingsen B. Metformin monotherapy for adults with type 2 diabetes mellitus. Cochrane Database Syst Rev. 2020;6(6):CD012906.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

